Author's response to reviews

Title: The GRONORUN 2 study: effectiveness of a preconditioning program on preventing running related injuries in novice runners. The design of a randomized controlled trial.

Authors:

   Steef W Bredeweg (s.w.bredeweg@sport.umcg.nl)
   Sjouke Zijlstra (s.zijlstra@sport.umcg.nl)
   Ida Buist (i.buist@sport.umcg.nl)

Version: 3 Date: 19 August 2010

Author's response to reviews: see over
Author's response to reviews

Title: The GRONORUN 2 study: effectiveness of a preconditioning program on preventing running related injuries in novice runners. The design of a randomized controlled trial.

Authors:

Steef W Bredeweg (s.w.bredeweg@sport.umcg.nl)
Sjouke Zijlstra (s.zijlstra@sport.umcg.nl)
Ida Buist (i.buist@sport.umcg.nl)

Version: 2 Date: 13 August 2010

Author's response to reviews: see over
Dear editor,

I hereby submit the revisions based on your comments on the manuscript of The GRONORUN 2 study: effectiveness of a preconditioning program on preventing running related injuries in novice runners. The design of a randomized controlled trial for publication in BMC Musculoskeletal Disorders. I changed the format of the title page and included some additional information in the authors’ contributions and the acknowledgement sections. I made no changes in the manuscript text nor did I make changes in the tables, figures or additional file.

I hope these revisions will satisfy the editorial board to proceed further for publication. Please contact me if you have any questions regarding this manuscript.

Sincerely,

Steef Bredeweg, MD

Physician in Sports Medicine
Center for Sports Medicine
University Medical Center Groningen
Hanzeplein 1
9713 GZ Groningen
The Netherlands
+31 50 361 7700
s.w.bredeweg@sport.umcg.nl